Author’s response to reviews

Title: Higher Proliferation of Peritumoral Endothelial Cells to IL-6/sIL-6R than Tumoral Endothelial cells in Hepatocellular Carcinoma

Authors:

Peng-Yuan Zhuang (pyzhuang@gmail.com)
Jian-Dong Wang (wangjiandong@163.com)
Zhao-Hui Tang (tangzhaohui@aliyun.com)
Xue-Ping Zhou (xp05zhou@163.com)
Zhi-Wei Quan (guanziwei@gmail.com)
Ying-Bin Liu (ilmare@gmail.com)
Jun Shen (junshen_xh@163.com)

Version: 4
Date: 5 September 2015

Author’s response to reviews:

Editorial requests:

In the Authors’ contributions you write that several authors ‘carried out the data collection and patients’ follow up’. Please include an ethics statement for the part of the study using patient information and tissue.

Response:

Thank you for pointing out our error. We have made the necessary correction in the revised manuscript in paragraph 2 page 5 and marked the text in yellow. We also made correction in the part of authors’ contribution and it now reads: PYZ, JS, ZHT, JDW, XPZ, YBL, ZWQ performed the experiments and collected data as well as patient information (in paragraph 4 page 18 and marked the text in yellow).